Author’s response to reviews

Title: Androgen secreting steroid cell tumor of the ovary in a young lactating women with acute onset severe hyperandrogenism

Authors:

Altarf Gauhar (altahaji786@yahoo.com)
Shekhar Sharma (drshekharsharma@gmail.com)
Vijaykumar D K (dkvijaykumar@aims.amrita.edu)
Manoj Babu (mmanojbabur@aims.amrita.edu)
chithrathara C (chithrathara@aims.amrita.edu)

Version: 9 Date: 13 November 2007

Author’s response to reviews:

13th November, 2007

Dear Editor,

I have revised the article as per instructions. The point by point response to the concerns raised by the editorial board is as follows:

Concern 1. In Table 1: The authors are using the unit of ngm/ml for Testosterone, but in the text (Case Report section), they used ngm/dl: Which is the right one? The authors must reconcile this discrepancy:

Explanation: The use of ngm/dl in the text of case report section was a mistake. We have corrected that. Therefore, the right one is ngm/ml which is mentioned in the table and the text.

Concern 2. In the cover letter, the authors stated they added one co-author, but I noted 2 being added. Please clarify.

Explanation: It was again a mistake on my part. Although I added 2 co-authors namely, Dr. Chitrathara and Dr. Manoj Babu but I forgot to mention that in the cover letter. Therefore, kindly note that I have added 2 co-authors.

Looking forward to your early response.

Thanks,

Dr. Altaf Gauhar Haji